Author's response to reviews

Title: Comparison of the Effects of Moderate and Severe Hypercapnic Acidosis on Ventilation-Induced Lung Injury

Authors:

Wanchao Yang (dachao_1980@126.com)
Ziyong Yue (yueziyong@126.com)
Xiaoguang Cui (cuixiaoguang@hotmail.com)
Yueping Guo (gyp721110@yahoo.com.cn)
Lili Zhang (zhanglili_80@126.com)
Huacheng Zhou (zhouhuacheng@163.com)
Wenzhi Li (Wenzhili9@126.com)

Version: 6 Date: 4 March 2015

Author's response to reviews: see over
Dear Prof. Tom Rowles:

Firstly, we thanks for your consideration of our manuscript entitled “Comparison of the Effects of Moderate and Severe Hypercapnic Acidosis on Ventilation-Induced Lung Injury” by Yang et al., for publication in BMC Anesthesiology. According to your comment, we have revised our manuscript and Table 1, Figure 2.

Secondly, we feel that the presented data adds valuable and clinically relevant information to the question when CO₂ should be applied in upcoming clinical studies. The undersigned authors warrant that the article is original, and has not been published previously.

None of the authors has financial affiliation which could employ any conflict of interest.

Sincerely yours,

Wanchao Yang and Wenzhi Li MD,PhD
Department of Anesthesiology, Second Affiliated Hospital of Harbin Medical University, Harbin, China.